Author’s response to reviews

Title: Survey of the Knowledge, Attitude and Practice of Nigerian Surgery Trainees to HIV-infected persons and AIDS patients

Authors:

Clement A Adebamowo (cadebamo@hsph.harvard.edu)
Emma R Ezeome (ezeome@infoweb.abs.net)
Johnson A Ajuwon (uchmed@skannet.com)
Temidayo O Ogundiran (toogundiran@yahoo.co.uk)

Version: 6 Date: 30 Aug 2002

The previously submitted version retained the changes that were made in MS Word track changes facility. A resubmission is therefore being made.